

# G OPEN ACCESS

**Citation:** Walters AGB, Gamble GD, Crowther CA, Dalziel SR, Eagleton CL, McKinlay CJD, et al. (2024) Cardiovascular outcomes 50 years after antenatal exposure to betamethasone: Follow-up of a randomised double-blind, placebo-controlled trial. PLoS Med 21(4): e1004378. https://doi.org/ 10.1371/journal.pmed.1004378

Academic Editor: Gordon C Smith, Cambridge University, UNITED KINGDOM

Received: October 15, 2023

Accepted: March 18, 2024

Published: April 1, 2024

**Copyright:** © 2024 Walters et al. This is an open access article distributed under the terms of the Creative Commons Attribution License, which permits unrestricted use, distribution, and reproduction in any medium, provided the original author and source are credited.

Data Availability Statement: De-identified participant data will be available to researchers who provide a methodologically sound proposal with appropriate ethical approval, where necessary, and following approval of the proposal by the Data Access Committee of the Liggins Institute. Data requestors will be required to sign a data access agreement before data are released. Request for access to data can be made to the Maternal and Perinatal Research Hub at the Liggins Institute, RESEARCH ARTICLE

# Cardiovascular outcomes 50 years after antenatal exposure to betamethasone: Follow-up of a randomised double-blind, placebo-controlled trial

Anthony G. B. Walters<sup>1</sup>, Greg D. Gamble<sup>1</sup>, Caroline A. Crowther<sup>1</sup>, Stuart R. Dalziel<sup>2,3</sup>, Carl L. Eagleton<sup>1</sup>, Christopher J. D. McKinlay<sup>2</sup>, Barry J. Milne<sup>4</sup>, Jane E. Harding<sup>1</sup>\*

1 Liggins Institute, University of Auckland, Auckland, New Zealand, 2 Department of Paediatrics, Child and Youth Health, University of Auckland, Auckland, New Zealand, 3 Department of Surgery, University of Auckland, Auckland, New Zealand, 4 Centre of Methods and Policy Application in Social Sciences, University of Auckland, Auckland, New Zealand

\* j.harding@auckland.ac.nz

# Abstract

# Background

Antenatal corticosteroids for women at risk of preterm birth reduce neonatal morbidity and mortality, but there is limited evidence regarding their effects on long-term health. This study assessed cardiovascular outcomes at 50 years after antenatal exposure to corticosteroids.

# Methods and findings

We assessed the adult offspring of women who participated in the first randomised, doubleblind, placebo-controlled trial of antenatal betamethasone for the prevention of neonatal respiratory distress syndrome (RDS) (1969 to 1974). The first 717 mothers received 2 intramuscular injections of 12 mg betamethasone or placebo 24 h apart and the subsequent 398 received 2 injections of 24 mg betamethasone or equivalent volume of placebo. Follow-up included a health questionnaire and consent to access administrative data sources. The coprimary outcomes were the prevalence of cardiovascular risk factors (any of hypertension, hyperlipidaemia, diabetes mellitus, gestational diabetes mellitus, or prediabetes) and age at first major adverse cardiovascular event (MACE) (cardiovascular death, myocardial infarction, coronary revascularisation, stroke, admission for peripheral vascular disease, and admission for heart failure). Analyses were adjusted for gestational age at entry, sex, and clustering. Of 1,218 infants born to 1,115 mothers, we followed up 424 (46% of survivors; 212 [50%] female) at mean (standard deviation) age 49.3 (1.0) years. There were no differences between those exposed to betamethasone or placebo for cardiovascular risk factors (159/229 [69.4%] versus 131/195 [67.2%]; adjusted relative risk 1.02, 95% confidence interval [CI] [0.89, 1.18;]; p = 0.735) or age at first MACE (adjusted hazard ratio 0.58, 95% CI [0.23, 1.49]; p = 0.261). There were also no differences in the components of these composite outcomes or in any of the other secondary outcomes. Key limitations were follow-up rate and lack of in-person assessments.

University of Auckland (researchhub@auckland.ac. nz).

Funding: This work was supported in part by: The Aotearoa Foundation (AW, Grant #9909494 awarded to JH). The Auckland Medical Research Foundation (AW, Grant number #1421003). https:// www.medicalresearch.org.nz/ Cure Kids New Zealand (SD is the Cure Kids Chair of Child Health Research). https://www.curekids.org.nz/ The Health Research Council of New Zealand (CC, GG, CM, BM, JH, Grant #19/690 awarded to JH). https://www.hrc.govt.nz/ The funders had no role in study design, data collection and analysis, decision to publish, or preparation of the manuscript.

**Competing interests:** The authors have declared that no competing interests exist.

Abbreviations: aMD, adjusted mean difference; AST, Auckland Steroid Trial; BMI, body mass index; BP, blood pressure; CI, confidence interval; DoHaD, developmental origins of health and disease; GLM, generalised linear mixed; HR, hazard ratio; LDL, low-density lipoprotein; MACE, major adverse cardiovascular event; MD, mean difference; NHI, National Health Index; RDS, respiratory distress syndrome.

## Conclusions

There is no evidence that antenatal corticosteroids increase the prevalence of cardiovascular risk factors or incidence of cardiovascular events up to 50 years of age. Established benefits of antenatal corticosteroids are not outweighed by an increase in adult cardiovascular disease.

# Author summary

## Why was this study done?

- Antenatal corticosteroids (ANC) are given to pregnant persons at risk of preterm birth to reduce the risk of death and breathing difficulty for their offspring.
- Animal studies have shown later adverse effects of ANC on offspring, particularly on their cardiovascular health in adulthood.
- There is little robust evidence about long-term effects in humans.
- This study aimed to provide evidence about the long-term effects of ANC on offspring cardiovascular health in adulthood.

## What did the researchers do and find?

- We assessed participants at 50 years of age who had been exposed to ANC or placebo during the first randomised clinical trial of this treatment.
- We compared cardiovascular disease and cardiovascular risk factors between those exposed to ANC and placebo in 424 participants (46% of those still alive), using a questionnaire and routinely collected administrative data.
- We found no differences in cardiovascular outcomes between those exposed to ANC and those who were not.

# What do these findings mean?

- There is no evidence that ANC has adverse effects on the cardiovascular health outcomes of offspring up to 50 years of age.
- The main limitation was the follow-up rate of less than 50% and the use of health questionnaires and routinely collected data rather than in-person assessments.

# Introduction

Antenatal corticosteroids are recommended for women at risk of preterm birth between 24 and 34 weeks' gestation [1,2] and have proven benefit for reducing neonatal morbidity and mortality and potentially reducing neurodevelopmental delay in childhood [3].

However, there are concerns that in utero exposure to corticosteroids could have long-term adverse effects [4]. The developmental origins of health and disease (DoHaD) hypothesis proposes that events in early life may influence long-term health [5], and one key proposed mechanism is exposure to excess corticosteroids during critical periods of development resulting in persisting changes in hypothalamic–pituitary–adrenal axis function [6,7]. In animal studies across a range of species, antenatal exposure to corticosteroids has resulted in altered cardio-vascular function including elevated blood pressure (BP) and sympathetic nerve activity, changes in hepatic metabolism, glucose homeostasis, and insulin resistance [7–14]. Human observational data have variably reported either higher or unchanged BP, although most studies assessed outcomes in adolescence or early adulthood [15–17]. A small number of observational studies of those born at very low birthweight have reported no association between antenatal corticosteroid exposure and glycated haemoglobin, plasma low-density lipoprotein (LDL) cholesterol, plasma triglycerides, or body mass index (BMI) [18,19], but may be confounded by the indication for treatment or health care disparities.

A 2020 Cochrane systematic review identified only 2 randomised trials of antenatal corticosteroids that have assessed offspring outcomes in early adulthood and noted the need for further follow-up studies to investigate effects in adulthood [3]. One study assessed 81 survivors at 20 years and found lower systolic BP in corticosteroid exposed participants but no effect on diastolic BP, anthropometry, or medical history [20]. The second assessed 534 30-year-old offspring of participants in the Auckland Steroid Trial (AST), the first and one of the largest randomised controlled trials of antenatal corticosteroids for women at risk of preterm birth, undertaken December 1969 to February 1974 [21,22]. Cardiovascular risk factors were not different between groups, including BP and lipid profile, but plasma insulin at 30 min after a 75 gram oral glucose tolerance test was higher in the betamethasone-exposed group, potentially suggesting mild insulin resistance [23]. However, the prevalence of cardiovascular risk factors and events is low at this age, and follow-up at older ages is required to assess clinically relevant outcomes.

In addition, although antenatal corticosteroids reduce neonatal respiratory distress syndrome (RDS), animal studies have reported that they interrupt lung development, leading to concerns that this could increase risk of emphysema in later life [8]. Previous follow-up of the AST indicated no differences in lung function tests between treatment groups at 30 years of age, but the prevalence of chronic obstructive airways disease is also low at this age [24].

We therefore assessed the offspring of participants in the AST at 50 years of age to investigate the long-term effects of antenatal exposure to corticosteroids on cardiovascular, metabolic, and respiratory outcomes.

#### Methods

#### **Ethics statement**

Ethics approval was obtained from the Northern A Health and Disability Ethics Committee on behalf of all New Zealand regional ethics committees. Written informed consent was obtained from each participant. A waiver of consent for accessing administrative health records was approved for potential participants who had died before the current follow-up study.

#### The Auckland Steroid Trial

The AST has been described elsewhere [21,22]. A double-blind, randomised, placebo-controlled trial of antenatal corticosteroids for preterm birth was performed at National Women's Hospital, Auckland, New Zealand between December 1969 and February 1974. All women expected to deliver between 24 and less than 37 weeks' gestation were eligible unless immediate delivery was indicated, or corticosteroids were contraindicated. Women provided oral informed consent for participation. Women were randomised to receive either betamethasone 12 mg intramuscularly (6 mg of short-acting betamethasone phosphate and 6 mg long-acting betamethasone acetate), repeated at 24 h, or an identical-appearing control (standard treatment dose). For planned preterm birth, the first injection was given 72 h before induction or cesarean delivery. The control treatment contained 6 mg cortisone acetate which has glucocorticoid potency one-seventieth of the active treatment and an identical appearance. As the initial findings indicated a reduction but not elimination of RDS in the treatment group, the dose of the betamethasone and control treatments were both doubled (twice standard treatment dose) in October 1972, after 717 women (769 infants) had been enrolled, in an attempt to further reduce rates of RDS [21]. For women in preterm labour, tocolytic therapy was administered to attempt to delay birth for 48 to 72 h from the first study injection, unless there were signs of fetal distress or amnionitis. A total of 1,115 women were enrolled and gave birth to 1,218 infants.

**Randomisation and masking.** A random-number table was used to generate the randomisation sequence. This was performed by the chief pharmacist who held the randomisation key. The pharmacy provided numbered, identical drug ampoules that contained either corticosteroids or the control treatment. Women received an intramuscular injection of the allocated treatment, followed by a further intramuscular injection of the same type after 24 h if delivery had not occurred. Participants were blinded to allocation from randomisation up to 30-year follow-up, after which 183 participants (83 in the betamethasone group and 100 in the placebo group) who wished to know their treatment allocation were informed by letter. Investigators and all study staff for the 50-year follow-up remained blinded to treatment allocation.

#### 50-year follow-up

**Participants.** All surviving offspring of women who took part in the AST were eligible to participate. Potential participants were traced using details recorded in prior follow-up studies, records from the AST (date of birth, mother's name, birthweight, gestation, and location at birth) and National Women's Hospital records. Additional contact information was sought using the National Health Index (NHI, a unique identifier assigned to every individual interacting with the New Zealand health system) database and associated demographic details, electoral rolls, and if required, internet and social media searches.

**Procedures.** Participants were asked to complete a questionnaire, either online or in hard copy, which was based on the New Zealand health survey 2019/2020 [25] and the International Primary care Airways Guidelines (IPAG) questionnaire [26]. They were also asked to consent for the study team to access administrative datasets managed by the New Zealand Ministry of Health [National Minimum Dataset (NMDS; hospital admissions since 1988), National Non-admitted Patients Collection (NNPaC; emergency department and outpatient visits since 2006), Mortality dataset (MORT; national mortality records since 1970), Pharmaceutical dataset (PHARM; national records of community dispensing of pharmaceuticals funded by the New Zealand government since 1992)] and TestSafe (laboratory tests data for Auckland and Northland regions which include 36% of New Zealand's population and New Zealand's largest city where all participants were born, since 2010).

This study is reported as per the Strengthening the Reporting of Observational Studies in Epidemiology (STROBE) guideline (<u>S1 Checklist</u>).

**Outcomes.** Outcomes used data from the questionnaire and administrative datasets. Where these were inconsistent, or only a single data source was available, all available data were reviewed by a panel of 5 clinicians (unaware of treatment allocation) to adjudicate the outcome. The co-primary outcomes were a composite of cardiovascular risk factors (any of diabetes mellitus [questionnaire, NMDS, PHARM, NNPaC, and TestSafe data], prediabetes [questionnaire and TestSafe data], gestational diabetes mellitus [questionnaire, NMDS, PHARM, and TestSafe data], hypertension [questionnaire, NMDS, and PHARM data], or hyperlipidaemia [questionnaire, PHARM, and TestSafe data]) and age at first MACE in those alive at 28 days after birth (hospital admission in the NMDS for myocardial infarction or coronary revascularisation, peripheral vascular disease [acute limb ischemia or revascularisation], stroke, heart failure) or cardiovascular death (mortality data).

Secondary outcomes include components of the primary composite outcomes; a hierarchical, 6 step, unmatched, win ratio [27] (time to death after randomisation, time to first MACE, diagnosis of diabetes mellitus, number of admissions to hospital with respiratory illness as primary reason for admission, self-reported general health and time in hospital per 10 years alive after 1988); death from any cause; ischemic heart disease; stroke; peripheral vascular disease; overweight or obesity and BMI; composite of obstructive airways disease (at least 1 of: selfreported diagnosis of asthma, self-reported diagnosis of chronic obstructive airways disease [COPD], hospital admission with asthma, hospital admission with COPD, prescription of pharmaceuticals for asthma or COPD, International Primary care Airways Group questionnaire score >19.5 [26]). Tertiary, exploratory outcomes included atrial fibrillation, heart failure diagnosis in those alive at follow-up, diagnosis of type 2 diabetes mellitus and type 1 diabetes mellitus in those alive at follow-up. Outcome definitions and data sources are available in S2 Appendix.

#### Statistical analysis

The statistical analysis plan was prepared before completion of data collection or any analyses (S1 Statistical Analysis Plan). All analyses were performed using SAS version 9.4 (SAS Institute, Cary, North Carolina, United States of America) and an intention-to-treat approach with participants analysed according to the initial treatment group to which their mother was allocated. Denominators are number of participants who consented to follow up with data available for the outcome, unless otherwise specified. There was no imputation for missing data, and participants for whom data were missing for a specific outcome were excluded from that analysis.

The sample size was limited by the number of surviving participants. For the co-primary composite outcome of cardiovascular risk factors, a total sample size of 420 would allow for detection of a 15% increase in the proportion with the outcome (relative risk (RR) = 1.33) with 90% power ( $\beta$  = 0.10,  $\alpha$  = 0.05), assuming a baseline prevalence of 45% in the placebo group (estimated from the New Zealand Health Survey 2020/2021 and published data on diabetes mellitus and prediabetes at age 45 to 55 years) [28,29]. For the co-primary outcome of age at first MACE, a total sample size of 420 would allow for detection of a hazard ratio (HR) of at least 0.33 with 90% power ( $\beta$  = 0.10,  $\alpha$  = 0.05), assuming a baseline prevalence of 8% in the placebo group (estimated from the New Zealand Health Survey 2020/2021 for age 45 to 55 years and adjusted for increased risk after preterm birth [28,30]).

Binary outcomes were compared between randomised groups using generalised linear mixed (GLM) modelling (binomial distribution, log-link function) to estimate RR and 95% confidence intervals (CIs) both unadjusted and adjusted for gestational age at randomisation and fetal sex. Adjustment for clustering of births from the same mother was performed using a random effects adjustment that approximated generalised estimation equations. Continuous outcomes were compared by GLM modelling (normal distribution, identity-link function) with and without the same adjustments and reported as mean difference (MD) or adjusted

mean difference (aMD) with 95% CIs. Significance testing was based on Type 3 fixed effects obtained from the GLM model for the binary and continuous outcomes and using joint testing for the Cox proportional hazards model. Each of the primary outcomes was tested at the 5% significance level. No adjustment to the significance level was made for secondary or tertiary outcomes as the study was a safety analysis and prioritised identifying important adverse effects at the risk of increasing likelihood of Type 1 error with multiple comparisons.

For time-to-event analyses, a Cox proportional hazards model was used, reported as an HR with 95% CIs both unadjusted and adjusted for gestational age at randomisation and fetal sex, after verifying the assumption of proportionality following visualisation of Kaplan–Meier curves. Censoring occurred at death or on completion of follow-up.

For the hierarchical win-ratio outcome, the 6 steps in the hierarchy were assessed sequentially, starting with time to death after randomisation, followed by time to MACE, diagnosis of diabetes mellitus, number of admissions to hospital for respiratory illness, self-reported general health, and time spent in hospital per 10 years alive. For each step, participants in the treatment and control group were compared in all possible pairs to determine "wins," "losses," and "ties." Pairwise comparisons with ties or missing data at any step were assessed at the next sequential step in the hierarchy. The analysis then determined the ratio of the probability of wins and losses for the treatment. The resulting odds ratio was reported with 95% CIs [27].

Prespecified subgroup analyses included: the planned study dose (standard versus twice standard dose), multiple pregnancy (singleton versus multiple), infant sex, tocolytic (ethanol versus salbutamol versus none/unknown), and the reason for risk of preterm birth. Interaction testing for subgroup analyses used the same model as the primary analysis, with the interaction between the subgroup and treatment group as an additional covariate. Subgroup analyses did not adjust for clustering to avoid overspecification of the model and failure of convergence. A post hoc subgroup analysis, requested in peer review, assessed the interaction between treatment group and gestational age at birth as both a categorical and continuous variable. Prespecified sensitivity analyses excluded participants who were randomised more than once in the same pregnancy (per protocol analysis), included additional covariates in the statistical models that were likely to be related to the outcomes of interest (BMI, sex, current socioeconomic status, gestation at birth, and birthweight z-score), excluded the second child for women who participated in more than 1 pregnancy, excluded participants for whom the reason for risk of preterm birth was listed as diabetes mellitus and participants for whom there was a known family history of diabetes mellitus (diabetes outcomes only), and excluded stillbirths and neonates who died before 28 days of age (hierarchical win ratio outcome only). We calculated birthweight z-scores using normative data from Thompson and colleagues [31].

#### Results

In the AST, 1,218 infants were born to 1,115 women and 987 survived to 28 days (Fig 1). Between January 2020 and May 2022, 424 adult survivors were recruited to this follow-up study (46% of those presumed alive), of whom 405 (96%) both completed a questionnaire and consented to data linkage, 9 (2%) only completed a questionnaire and 10 (2%) only consented to data linkage.

Those eligible who did and did not participate had similar baseline characteristics, though participants were more likely to be female (50.0% versus 40.8%) and were less likely to be born at term (29.0% versus 40.2%) (Table 1). Those who died prior to 50-year follow-up had a lower gestational age at trial entry and at birth and were more likely to have had RDS and a low Apgar score. Baseline characteristics of participants whose mothers were randomised to



Fig 1. Trial flow diagram for the Auckland Steroid Trial 50-year follow-up. \*Number of participants assessed for eligibility not available in original trial 1969–1974 records. †Included participants successfully traced but a response was not obtained. MACE, major adverse cardiovascular event. ‡Includes participants assessed at 50-year follow-up and participants deceased after 28 days of age for whom administrative data was available. Denominator is participants alive at 28 days (493 in betamethasone and 494 in placebo group).

https://doi.org/10.1371/journal.pmed.1004378.g001

betamethasone were similar to those randomised to placebo, except the betamethasone group had a lower proportion of females (45.9% versus 54.9%) (Table 2) [32].

The co-primary composite outcome of cardiovascular risk factors occurred in 159/229 (69.4%) of the betamethasone group and 131/195 (67.2%) of the placebo group (GLM aRR 1.02, 95% CI [0.89, 1.18]; p = 0.735). The co-primary outcome of age at first MACE was (median, 5th, 95th centile) 41.8 years (20.5, 48.5) in the betamethasone group and 44.0 years (24.2, 50.3) in the placebo group (Cox aHR 0.58, 95% CI [0.23, 1.49]; p = 0.261), although the proportion of participants with any event was low (1.4% betamethasone group; 2.4% placebo group) (Table 3).

All of the components of the composite of cardiovascular risk factors outcome were similar between groups (Table 3). Age at first event for the components of the MACE outcome did not differ between treatment groups (Table 3).

The 6-step, unmatched win ratio was 1.18 (95% CI [0.98, 1.42]; p = 0.088), with a total of 110,274 wins (representing a better outcome) in the betamethasone group and 93,728 wins in the placebo group (Table 3). Each of the outcomes making up the 6 steps of the win ratio did not differ between groups (Table 3). The number of wins and losses at each step are available in Fig A in S2 Appendix.

Treatment groups did not differ in self-reported height; GLM aMD 1.3 centimetres (cm) (95% CI [-0.4, 3.1]; p = 0.121), weight (aMD 3.0 kilograms (kg), 95% CI [-1.3, 7.3]; p = 0.160), and BMI (aMD 0.8 kg/m<sup>2</sup>, 95% CI [-0.7, 2.3]; p = 0.259), nor the proportion of overweight or obesity (aRR 1.05, 95% CI [0.92, 1.21]; p = 0.432) (Table 3).

#### Table 1. Baseline characteristics of those eligible who did and did not participate.

	Participated N = 424	Did not participate N = 493	Deceased N = 301
Randomised to antenatal corticosteroid	229 (54.0%)	233 (47.3%)	139 (46.2%)
Female	212 (50.0%)	201 (40.8%)	128 (42.5%)
Gestational age at entry, weeks, median (5th, 95th centile)	33.1 (27.6, 36.0)	33.4 (27.5, 37.0)	29.6 (24.0, 36.0)
Gestational age at delivery, weeks, median (5th, 95th centile)	35.0 (29.3, 40.6)	36.0 (30.7, 41.0)	31.3 (24.0, 40.0)
Multiple pregnancy			
Singleton	370 (87.3%)	437 (88.6%)	261 (86.7%)
Multiple	54 (12.7%)	56 (11.4%)	40 (13.3%)
Unplanned premature labour	354 (83.5%)	435 (88.2%)	247 (82.1%)
Planned delivery			
Hypertension-oedema-proteinuria syndromes	39 (9.2%)	27 (5.5%)	29 (9.6%)
Rhesus isoimmunisation	18 (4.3%)	14 (2.8%)	14 (4.7%)
Placenta previa	8 (1.9%)	11 (2.2%)	6 (2.0%)
Diabetes mellitus	5 (1.2%)	6 (1.2%)	5 (1.7%)
Mode of delivery			
Normal vaginal delivery	193 (45.5%)	225 (45.6%)	168 (55.8%)
Instrumental delivery	184 (43.4%)	210 (42.6%)	112 (37.2%)
Cesarean section	47 (11.1%)	58 (11.8%)	21 (7.0%)
Tocolytic used			
None	138 (32.6%)	144 (29.2%)	85 (28.2%)
Ethanol	67 (15.8%)	67 (13.6%)	50 (16.6%)
Salbutamol	207 (48.8%)	272 (55.2%)	160 (53.2%)
Both	12 (2.8%)	10 (2.0%)	6 (2.0%)
Treatment dose of betamethasone or placebo			
Standard dose (2 doses of 12 mg)	268 (63.2%)	298 (60.5%)	203 (67.4%)
Twice standard dose (2 doses of 24 mg)	156 (36.8%)	195 (39.6%)	98 (32.6%)
Term delivery	123 (29.0%)	198 (40.2%)	34 (11.3%)
Birthweight, grams, mean (SD)	2,317 (747)	2,483 (717)	1,659 (833)
Birthweight Z score, mean (SD)	-0.36 (0.98)	-0.42 (0.96)	-0.31 (1.28)
5-min Apgar score <7	53 (12.6%)	59 (12.1%)	92 (46.0%)
RDS	37 (8.7%)	30 (6.1%)	75 (24.9%)

mg, milligrams; RDS, respiratory distress syndrome; SD, standard deviation.

https://doi.org/10.1371/journal.pmed.1004378.t001

All-cause mortality from randomisation to 50-year follow-up was 139/601 (23.1%) in the betamethasone group and 165/617 (26.7%) in the placebo group (GLM aRR 0.89, 95% CI [0.74, 1.08]; p = 0.236). Causes of death more than 28 days after birth were similar between groups (Table 4).

The composite outcome for ischemic heart disease and stroke did not differ between groups (Table 3). No participants met the definition for peripheral vascular disease.

The proportion with the composite of obstructive airways disease did not differ between the betamethasone group (77/229, 41.5%) and the placebo group (66/195, 38.5%) (GLM aRR 1.01, 95% CI [0.76, 1.35]; p = 0.930). The components of this composite also did not differ between groups (Table 3).

The prespecified subgroup analyses did not reveal any interactions between treatment effect and the dose of antenatal corticosteroid administered, multiple pregnancy, infant sex, tocolytic used or reason for risk of preterm birth, although the number of participants and outcomes in

	Betamethasone N = 229	Placebo N = 195
Maternal characteristics		
Gestational age at entry, weeks, median (5th, 95th centile)	32.9 (27.6, 36.0)	33.4 (27.4, 36.0)
Gestational age at delivery, weeks, median (5th, 95th centile)	34.9 (29.0, 40.6)	35.0 (30.0, 40.9)
Multiple pregnancy		
Singleton	195 (85.2%)	175 (89.7%)
Multiple	34 (14.9%)	20 (10.3%)
Unplanned premature labour	193 (84.3%)	161 (82.6%)
Planned delivery		
Hypertension-oedema-proteinuria syndromes	20 (8.7%)	19 (9.7%)
Rhesus isoimmunisation	10 (4.4%)	8 (4.1%)
Placenta previa	4 (1.8%)	4 (2.1%)
Diabetes mellitus	2 (0.9%)	3 (1.5%)
Mode of delivery		
Normal vaginal delivery	102 (44.5%)	91 (46.7%)
Instrumental delivery	104 (45.4%)	80 (41.0%)
Cesarean section	23 (10.0%)	24 (12.3%)
Tocolytic used		
None	73 (31.9%)	65 (33.3%)
Ethanol	38 (16.6%)	29 (14.9%)
Salbutamol	111 (48.5%)	96 (49.2%)
Both	7 (3.1%)	5 (2.6%)
Treatment dose of betamethasone or placebo		
Standard dose (2 doses of 12 mg)	145 (63.3%)	123 (63.1%)
Twice standard dose (2 doses of 24 mg)	84 (36.7%)	72 (36.9%)
Multiple randomisation in same pregnancy†	6 (2.6%)	4 (2.1%)
Multiple randomisation in different pregnancies	6 (2.6%)	4 (2.1%)
Neonatal characteristics		
Female	105 (45.9%)	107 (54.9%)
Term delivery	63 (27.5%)	60 (30.8%)
Birthweight, grams, mean, (SD)	2,313 (776)	2,322 (714)
Birthweight Z score, mean (SD)	-0.32 (0.93)	-0.42 (1.03)
5-min Apgar score <7	31 (13.5%)	22 (11.5%)
RDS	15 (6.6%)	22 (11.3%)
Adult characteristics		
Age at entry to follow-up study, years, mean (SD)	49.3 (1.0)	49.3 (1.0)
Age at most recent administrative data, years, mean (SD)	50.4 (1.0)	50.4 (1.0)
Ethnicity*		
Māori	43 (18.8%)	54 (27.7%)
European	164 (71.6%)	132 (67.7%)
Pacific	15 (6.6%)	5 (2.6%)
Asian	0 (0%)	1 (0.5%)
Other	1 (0.4%)	0 (0%)
Unknown	6 (2.6%)	3 (1.5%)
Smoking status	. ,	. ,
Non-smoker	124 (55.9%)	98 (51.9%)
Previously smoked	64 (28.8%)	69 (36.5%)

Table 2.	Baseline and adult characteristics of participants whose mothers were randomised to betamethasone and
placebo.	

(Continued)

Table 2. (Continued)

	Betamethasone N = 229	Placebo N = 195
Currently smokes	34 (15.3%)	22 (11.6%)
Unblinding at 30-year follow-up	68 (29.7%)	68 (34.9%)

\*Self-reported ethnicity. If multiple ethnicities were recorded, ethnicity was prioritised as Māori, Pacific, Asian, Other, and European [32].

<sup>†</sup>Three participants had received betamethasone on first randomisation followed by placebo when randomised later in the pregnancy, 3 received placebo followed by betamethasone, 3 received betamethasone on both occasions, and 1 received placebo on both occasions.

mg, milligrams; RDS, respiratory distress syndrome; SD, standard deviation.

https://doi.org/10.1371/journal.pmed.1004378.t002

subgroups was very low in some cases and findings should be interpreted with caution (Table C in <u>S2 Appendix</u>). The win ratio did not differ markedly in sensitivity analyses excluding stillbirths or excluding deaths before 28 days (Table D in <u>S2 Appendix</u>). Other sensitivity analyses, including adjustment for gestation length, also did not markedly alter the findings for the primary and secondary outcomes (Table E in <u>S2 Appendix</u>). There was also no significant interaction between treatment effect and gestational age at delivery (post hoc analysis requested during peer review, Table F in <u>S2 Appendix</u>).

#### Discussion

We studied cardiovascular outcomes at 50 years in 424 offspring born to mothers who participated in the first, and one of the largest, randomised controlled trials of antenatal corticosteroids prior to preterm birth. We found no difference between treatment groups for the coprimary outcomes of composite of cardiovascular risk factors or age at first MACE, or for their components. The composite outcome of cardiovascular risk factors also included prediabetes and gestational diabetes mellitus, which are associated with cardiovascular disease [33,34]. We also found no difference in a win ratio, intended to provide a more global assessment of the impact of antenatal betamethasone, incorporating both early and late mortality, along with cardiovascular events, diabetes diagnosis, respiratory admissions, self-reported general health, and time in hospital. To our knowledge, these findings represent the longest follow-up of children with antenatal exposure to betamethasone and demonstrate that at 50 years, there is no evidence of an increase in cardiovascular risk factors or adverse cardiovascular events. Thus, the substantial neonatal benefits of antenatal corticosteroids for preterm birth do not appear to be outweighed by negative later consequences [1]. Moreover, antenatal corticosteroids administered prior to preterm birth reduce early life healthcare care costs [35], and our findings suggest that they are very likely to remain cost-effective through into adulthood.

Our results are consistent with the findings in the earlier, 30-year, follow-up of offspring of participants in this clinical trial, which showed no differences between groups in rates of hypertension, systolic or diastolic BP, diabetes mellitus, or fasting lipid concentrations [23]. It is also consistent with another study following up 81 participants of a randomised controlled trial that found participants exposed to antenatal betamethasone had a lower systolic BP at 20 years than those exposed to placebo, although laboratory testing for dyslipidaemia and diabetes mellitus was not undertaken [20]. A recent systematic review of cardiovascular function after antenatal corticosteroid exposure, based predominantly on observational data, also concluded that there was no association between antenatal corticosteroid exposure and offspring BP but

#### Table 3. Primary and secondary outcomes.

· · ·					
Outcome	Betamethasone	Placebo	Unadjusted RR, HR, win ratio, or MD (95% CI)	Adjusted* RR, HR, or MD (95% CI)	P value (adjusted*)
Cardiometabolic risk factor composite, <i>n</i> / <i>N</i>	159/229 (69.4%)	131/195 (67.2%)	RR 1.03 (0.91, 1.18)	RR 1.02 (0.89, 1.18)	0.735
Age at first MACE, years, median (5th, 95th centile)	41.8 (20.5, 48.5)	44.0 (24.2, 50.3)	HR 0.58 (0.23, 1.47)	HR 0.58 (0.23, 1.49)	0.261
Diabetes mellitus, prediabetes or gestational diabetes mellitus, $n/N$	48/229 (21.0%)	41/195 (21.0%)	RR 1.00 (0.69, 1.45)	RR 1.00 (0.67, 1.49)	0.981
Diabetes mellitus, <i>n</i> /N	15/229 (6.6%)	21/195 (10.8%)	RR 0.61 (0.32, 1.15)	RR 0.59 (0.29, 1.19)	0.129
Prediabetes, n/N	30/228 (13.2%)	18/194 (9.3%)	RR 1.42 (0.82, 2.47)	RR 1.42 (0.78, 2.59)	0.234
Gestational diabetes mellitus, n/N	7/105 (6.7%)	3/106 (2.8%)	RR 2.36 (0.62, 8.93)	RR 2.71 (0.45, 16.4)	0.214
Hypertension, <i>n</i> / <i>N</i>	72/228 (31.6%)	64/195 (32.8%)	RR 0.96 (0.73, 1.27)	RR 0.96 (0.72, 1.28)	0.758
Hyperlipidaemia, n/N	131/228 (57.5%)	112/194 (57.7%)	RR 1.00 (0.84, 1.17)	RR 0.98 (0.82, 1.17)	0.7880
At least 1 admission for MACE including cardiovascular death, <i>n</i> / <i>N</i>	7/493 (1.4%)	12/494 (2.4%)	RR 0.54 (0.22, 1.34)	RR 0.55 (0.22, 1.39)	0.201
Cardiovascular death, <i>n</i> / <i>N</i>	2/493 (0.4%)	2/494 (0.4%)	RR 1.00 (0.14, 7.10)	RR 0.97 (0.13, 7.16)	0.977
Age at cardiovascular death, years, median (5th, 95th centile)	33.1 (20.5, 45.8)	46.9 (46.0, 47.7)	HR 1.00 (0.14, 7.07)	HR 0.96 (0.13, 6.84)	0.965
At least 1 admission for myocardial infarction or coronary revascularisation, $n/N$	5/493 (1.0%)	4/494 (0.8%)	RR 1.25 (0.34, 4.64)	RR 1.29 (0.34, 4.93)	0.705
Age at first admission for myocardial infarction or coronary revascularisation, years, median (5th, 95th centile)	44.6 (41.4, 45.2)	40.9 (26.4, 47.1)	HR 1.25 (0.34, 4.66)	HR 1.29 (0.34, 4.83)	0.706
At least 1 admission for myocardial infarction, <i>n</i> / <i>N</i>	4/493 (0.8%)	2/494 (0.4%)	RR 2.00 (0.37, 10.91)	RR 2.09 (0.37, 11.78)	0.400
Age at first admission for myocardial infarction, years, median (5th, 95th centile)	43.7 (41.4, 45.2)	41.9 (37.6, 46.2)	HR 2.01 (0.37, 10.96)	HR 2.09 (0.38, 11.47)	0.397
At least 1 admission for coronary revascularisation, <i>n</i> / <i>N</i>	1/493 (0.2%)	4/494 (0.8%)	RR 0.25 (0.03, 2.24)	RR 0.28 (0.03, 2.63)	0.260
Age at first admission for coronary revascularisation, years, median (5th, 95th centile)	44.6 (44.6, 44.6)	40.9 (26.4, 47.1)	HR 0.25 (0.03, 2.23)	HR 0.28 (0.03, 2.51)	0.254
At least 1 admission for peripheral vascular disease, $n/N$	0/493 (0%)	0/494 (0%)	NE	NE	NE
Age at first admission for peripheral vascular disease, years, median (5th, 95th centile)	NA	NA	NE	NE	NE
At least 1 admission for stroke, <i>n</i> / <i>N</i>	1/493 (0.2%)	2/494 (0.4%)	RR 0.50 (0.05, 5.52)	RR 0.52 (0.04, 5.99)	0.592
Age at first admission for stroke, years, median (5th, 95th centile)	47.0 (47.0, 47.0)	48.9 (47.6, 50.3)	HR 0.50 (0.05, 5.52)	HR 0.52 (0.05, 5.73)	0.590
At least 1 admission for heart failure, <i>n</i> / <i>N</i>	3/493 (0.6%)	7/494 (1.4%)	RR 0.43 (0.11, 1.65)	RR 0.43 (0.11, 1.69)	0.221
Age at first heart failure admission, years, median (5th, 95th centile)	41.8 (35.7, 48.5)	40.3 (24.2, 47.1)	HR 0.43 (0.11, 1.65)	HR 0.43 (0.11, 1.65)	0.216
Death from any cause, <i>n</i> / <i>N</i>	139/601 (23.1%)	165/617 (26.7%)	RR 0.86 (0.71, 1.05)	RR 0.89 (0.74, 1.08)	0.236
Win ratio hierarchical outcome	Treatment wins = 110,274	Placebo wins = 93,728	Win ratio† 1.18 (0.98, 1.42)	NA	0.088
Time to death after randomisation, days, median (5th, 95th centile)	10.5 (0.4, 12,850)	4.0 (0.3, 16,902)	HR 0.84 (0.67, 1.05)	HR 0.82 (0.65, 1.03)	0.084
Time to MACE after randomisation (excluding cardiovascular death), days, median, (5th, 95th centile)	15,351 (13,132, 17,718)	16,064 (8,874, 18,444)	HR 0.54 (0.20, 1.46)	HR 0.55 (0.20, 1.50)	0.245
Number of admissions to hospital with respiratory illness as primary reason for admission since 1988, mean (SD)	0.15 (0.67)	0.34 (1.91)	MD -0.19 (-0.44, 0.07)	MD -0.17 (-0.46, 0.11)	0.213
Self-reported general health, <i>n</i> / <i>N</i>					0.579
Excellent	31 (13.9%)	35 (18.3%)	NA	NA	
Very good	74 (33.2%)	67 (35.1%)			
Good	81 (36.3%)	65 (34.0%)			
Fair	29 (13.0%)	20 (10.5%)			
Poor	8 (3.6%)	4 (2.1%)			

(Continued)

#### Table 3. (Continued)

Outcome	Betamethasone	Placebo	Unadjusted RR, HR, win ratio, or MD (95% CI)	Adjusted* RR, HR, or MD (95% CI)	P value (adjusted*)
Self-reported general health fair/poor, <i>n</i> /N	37/223 (16.6%)	24/191 (12.6%)	RR 1.32 (0.82, 2.13)	RR 1.37 (0.82, 2.29)	0.216
Time in hospital after 1988, days per 10 years alive, median (5th, 95th centile)	2.0 (0.0, 23.0)	2.6 (0.0, 31.6)	MD -6.5 (-15.2, 2.1)	MD -6.2 (-15.7, 3.2)	0.181
Ischemic heart disease, <i>n</i> /N	5/229 (2.2%)	9/195 (4.6%)	RR 0.47 (0.16, 1.39)	RR 0.46 (0.14, 1.49)	0.181
Stroke, <i>n/N</i>	1/229 (0.4%)	3/195 (1.5%)	RR 0.28 (0.03, 2.72)	RR 0.30 (0.03, 3.51)	0.316
Peripheral vascular disease, n/N	0/229 (0%)	0/195 (0%)	NE	NE	NE
Weight, kg, mean (SD)	88.8 (1.4)	84.9 (1.6)	MD 4.0 (-0.2, 8.1)	MD 3.0 (-1.3, 7.3)	0.160
Height, cm, mean (SD)	172.8 (0.7)	170.2 (0.8)	MD 2.5 (0.5, 4.6)	MD 1.3 (-0.4, 3.1)	0.121
BMI, kg/m <sup>2</sup> , mean (SD)	29.7 (6.4)	29.0 (6.2)	MD 0.7 (-0.6, 2.0)	MD 0.8 (-0.7, 2.3)	0.259
Overweight or obesity, <i>n</i> / <i>N</i>	152/198 (76.8%)	114/160 (71.3%)	RR 1.08 (0.95, 1.22)	RR 1.05 (0.92, 1.21)	0.432
Overweight, <i>n/N</i>	73/198 (36.9%)	59/160 (36.9%)	RR 1.00 (0.76, 1.31)	RR 0.94 (0.70, 1.27)	0.678
Obesity, n/N	79/198 (39.9%)	55/160 (34.4%)	RR 1.16 (0.88, 1.53)	RR 1.17 (0.87, 1.58)	0.265
Obesity class 1 (BMI 30- <35 kg/m <sup>2</sup> ), <i>n</i> /N	45/160 (22.7%)	30/160 (18.8%)	RR 1.21 (0.8, 1.83)	RR 1.22 (0.78, 1.9)	0.350
Obesity class 2 (BMI 35- $<40 \text{ kg/m}^2$ ), $n/N$	19/198 (9.6%)	18/160 (11.3%)	RR 0.85 (0.46, 1.57)	RR 0.88 (0.46, 1.7)	0.678
Obesity class 3 (BMI $\geq$ 40 kg/m <sup>2</sup> ), <i>n</i> /N	15/198 (7.6%)	7/160 (4.4%)	RR 1.73 (0.72, 4.16)	RR 1.86 (0.70, 4.96)	0.191
Composite of obstructive airways disease, <i>n</i> / <i>N</i>	77/229 (41.5%)	66/195 (38.5%)	RR 0.99 (0.76, 1.30)	RR 1.01 (0.76, 1.35)	0.930
Asthma, n/N	76/229 (33.2%)	65/195 (33.3%)	RR 1.00 (0.76, 1.31)	RR 1.01 (0.75, 1.34)	0.963
COPD, n/N	4/229 (1.8%)	4/195 (2.1%)	RR 0.85 (0.21, 3.37)	RR 1.10 (0.25, 4.75)	0.896
IPAG COPD questionnaire score >19.5, <i>n</i> /N	3/179 (1.7%)	2/147 (1.4%)	RR 1.23 (0.21, 7.32)	RR 1.22 (0.16, 9.20)	0.828
At least 1 admission for asthma or COPD, <i>n</i> / <i>N</i>	20/223 (9.0%)	16/188 (8.5%)	RR 1.05 (0.56, 1.98)	RR 1.12 (0.56, 2.23)	0.736
Prescriptions of pharmaceuticals for asthma or COPD, $n/N$	69/219 (31.5%)	47/182 (25.8%)	RR 1.22 (0.89, 1.67)	RR 1.29 (0.92, 1.82)	0.128

<sup>†</sup>Hierarchical win-odds ratio: 6-step, unmatched, win ratio hierarchy including time to death after randomisation; time to first MACE; diagnosis of diabetes mellitus; number of admissions to hospital with respiratory illness as primary reason for admission; self-reported general health; time in hospital per 10 years alive after 1988. \*Analyses adjusted for sex, gestational age at trial entry and for clustering.

BMI, body mass index; CI, confidence interval; cm, centimetres; COPD, chronic obstructive pulmonary disease; HR, hazard ratio; IPAG, International primary care airways group; MACE, major adverse cardiovascular event; MD, mean difference; NA, not applicable; NE, not estimable; RR, relative risk.

https://doi.org/10.1371/journal.pmed.1004378.t003

#### Table 4. Causes of death after 28 days of age.

Cause of death	Betamethasone	Placebo*	Total
Cardiovascular	2 (6%)	2 (5%)	4 (6%)
Cerebral palsy	1 (3%)	1 (2%)	2 (3%)
Complications of diabetes mellitus	0 (0%)	2 (5%)	2 (3%)
Congenital abnormalities	2 (6%)	3 (7%)	5 (7%)
Infection	5 (16%)	8 (19%)	13 (18%)
Malignancy	4 (13%)	7 (17%)	11 (15%)
Sudden unexpected death in infancy	3 (10%)	1 (2%)	5 (7%)
Suicide	3 (10%)	2 (5%)	5 (7%)
Trauma/accidental death	5 (16%)	6 (14%)	11 (15%)
Other	2 (6%)	3 (7%)	5 (7%)
Unknown	4 (13%)	7 (17%)	11 (15%)
Total	31 (100%)	42 (100%)	73 (100%)

\*Table includes 3 participants in the placebo group who died between completion of questionnaire and analysis of linked administrative data.

https://doi.org/10.1371/journal.pmed.1004378.t004

the evidence was too limited to draw firm conclusions about other aspects of cardiovascular function [17]. Concerns regarding premature cardiovascular disease after antenatal exposure to corticosteroids have followed the evidence of increased cardiovascular risk factors in animal models and human evidence for a potential role of the hypothalamic–pituitary–adrenal axis in cardiovascular disease pathophysiology [8–11,36]. Previous studies investigating the cardiovascular effects of antenatal corticosteroids, both follow-up of clinical trials and observational studies, have focused on secondary endpoints such as risk factors rather than adverse cardiovascular events which are clinically of greater importance [8,17,37,38]. Our findings of no difference in the prevalence of risk factors or of cardiovascular events suggest that any effect on later cardiometabolic health of exposure to a single course of antenatal corticosteroids is unlikely to be clinically important, at least up to 50 years.

There have been concerns that in utero exposure to antenatal corticosteroids may interrupt lung branching and alveolarization, potentially increasing the risk of later obstructive airways disease [8]. Consistent with this, increasing airway obstruction has been reported between 8 and 18 years in extremely preterm or low birthweight survivors, 73% of whom were exposed to antenatal corticosteroids [39]. We did not identify any difference between groups for the composite outcome of obstructive airways disease, although most participants with this outcome had asthma rather than COPD, and there were few extremely preterm infants. This finding is consistent with the lack of a difference in lung function, as assessed by spirometry, between betamethasone and placebo groups at 30 years in an earlier follow-up of participants in the AST, and between 6- and 8-year-old children exposed to single or repeat courses of antenatal corticosteroids in a more recent trial [24,40]. Although COPD is still uncommon at age 50, our findings suggest that premature COPD seems unlikely to be of clinical concern after antenatal corticosteroids.

Due to difficulties in predicting preterm birth, not all infants receiving antenatal corticosteroids will be born preterm, and recent studies have raised concerns that those born at term, who do not ultimately benefit from corticosteroid exposure may be at a small increased risk of adverse long-term outcomes compared to term-born controls who were not exposed to antenatal corticosteroids [41]. However, these cohort studies are inevitably confounded by the indications for use of antenatal corticosteroids and healthcare disparities. In the AST, where the control group had similar indications for the use of antenatal corticosteroids as those in the betamethasone group, just under one third of participants were born at term, similar to the proportion seen in current clinical practice [42]. We also found no significant interaction between treatment effect and gestational age at birth. Thus, the present study provide reassurance that use of antenatal corticosteroids for threatened or indicated preterm birth provides net clinical benefit in the short- and long-term, even though the prediction of timing of early birth remains imperfect.

Strengths of this study include that it assesses outcomes from a randomised trial, greatly reducing the risk of bias due to confounding. Observational studies on the long-term effects of antenatal corticosteroids are particularly difficult to interpret because of the confounding effects of healthcare disparities and the indication for treatment with antenatal corticosteroid.

We combined self-reported questionnaire data with information from administrative datasets to improve identification of outcomes by reducing the impact of recall bias, as selfreported data alone is known to underestimate prevalence of noncommunicable diseases such as hypertension, hyperlipidaemia, and diabetes mellitus [43,44]. However, the use of administrative data has important limitations, including the risk of outcome misclassification through miscoding at data collection, lack of specificity in outcome definitions (for example, through classifying diagnoses according to treatments), and the influence of healthcare access on likelihood of appearing in an administrative dataset. Nevertheless, there is no reason to believe that these biases differed between the randomised groups, although they could contribute bias towards a null finding.

There were a number of limitations of this study. First, the follow-up rate was relatively low (46% of those presumed to be alive, primarily driven by difficulty in tracing participants decades after the last study contact), which increases the risk of attrition bias. However, most baseline characteristics were similar between those assessed and those lost to follow-up, and there were few baseline differences between randomised groups. Further, additional adjustments and sensitivity analyses to explore these differences did not alter the findings. Second, we did not undertake in person assessment of BP and blood testing, for logistic reasons due to geographical dispersal of the cohort over 50 years, so the outcomes of hypertension, hyperlipidaemia, and diabetes may be underestimated. However, there is no reason to think this underestimate differed between randomised groups. Third, the overall event rate for the MACE was lower than anticipated, so our study had limited power to detect small differences between groups. However, given the substantial neonatal benefits of exposure to antenatal corticosteroids, small differences between groups in the primary outcomes at 50 years are unlikely to be clinically important.

Finally, there was a higher proportion of women in the placebo group than the betamethasone group, consistent with the known increased mortality of male preterm neonates, especially in the absence of antenatal corticosteroid treatment. This difference would be likely to favour the placebo group given the lower risk of cardiovascular disease in women. Adjustment for sex did not alter results and there was no evidence in subgroup analyses that sex influences the effect of betamethasone on primary or secondary outcomes. Thus, the imbalance in sex is unlikely to have affected our findings.

We found no evidence that the prevalence of cardiovascular risk factors or age at MACE is altered up to 50 years after antenatal exposure to betamethasone. Clinicians can have confidence that the established benefits of antenatal corticosteroids for preterm birth in reducing neonatal morbidity and mortality and probable benefits for childhood neurodevelopment are not outweighed by an increase in adult cardiovascular disease.

# Supporting information

S1 STROBE checklist. STROBE Statement—Checklist of items that should be included in reports of *cohort studies*.

(DOC)

S1 Statistical Analysis Plan. The AnteNatal Corticosteroids Health Outcomes Review (ANCHOR) Study: Auckland Steroid Trial follow-up. (PDF)

**S1** Appendix. Supplement to statistical analysis plan: Outcome definitions. (DOCX)

**S2 Appendix.** Table A. Tertiary outcomes. Fig A. Wins and losses for betamethasone at each step of the 6-step unmatched win ratio. Table B. Subgroup analyses for primary outcomes. Table C. Subgroup analyses for secondary outcomes. Table D. Sensitivity analysis for win ratio hierarchical outcome. Table E. Sensitivity analysis for primary and secondary outcomes. Table F. Post hoc subgroup analysis for primary outcomes and components by gestational age at delivery.

(DOCX)

#### Acknowledgments

We thank the mothers and their offspring who participated in the original trial and this follow-up study, without whom this work would not have been possible, and acknowledge the seminal contributions of the original trialists, the late Ross Howie and Mont Liggins, who generously made available the original data and documentation from the Auckland Steroid Trial. We also thank Trisha Meagher-Lundberg for study coordination and Coila Bevan for participant tracing.

#### **Author Contributions**

- **Conceptualization:** Caroline A. Crowther, Stuart R. Dalziel, Christopher J. D. McKinlay, Jane E. Harding.
- Data curation: Anthony G. B. Walters, Greg D. Gamble.
- Formal analysis: Anthony G. B. Walters, Greg D. Gamble.
- Investigation: Anthony G. B. Walters.
- Methodology: Anthony G. B. Walters, Greg D. Gamble, Caroline A. Crowther, Stuart R. Dalziel, Carl L. Eagleton, Christopher J. D. McKinlay, Barry J. Milne, Jane E. Harding.
- Supervision: Greg D. Gamble, Caroline A. Crowther, Stuart R. Dalziel, Carl L. Eagleton, Christopher J. D. McKinlay, Barry J. Milne, Jane E. Harding.
- Writing original draft: Anthony G. B. Walters, Greg D. Gamble, Caroline A. Crowther, Stuart R. Dalziel, Carl L. Eagleton, Christopher J. D. McKinlay, Barry J. Milne, Jane E. Harding.
- Writing review & editing: Anthony G. B. Walters, Greg D. Gamble, Caroline A. Crowther, Stuart R. Dalziel, Carl L. Eagleton, Christopher J. D. McKinlay, Barry J. Milne, Jane E. Harding.

#### References

- World Health Organization. WHO recommendations on antenatal corticosteroids for improving preterm birth outcomes. Geneva: World Health Organization; 2022. https://apps.who.int/iris/handle/10665/ 363131.
- National Collaborating Centre for Women's and Children's Health (UK). Preterm labour and birth. London: National Institute for Health and Care Excellence (UK); 2015. <u>http://www.ncbi.nlm.nih.gov/books/ NBK327571/</u>.
- McGoldrick E, Stewart F, Parker R, Dalziel SR. Antenatal corticosteroids for accelerating fetal lung maturation for women at risk of preterm birth. Cochrane Pregnancy and Childbirth Group, editor. Cochrane Database of Systematic Reviews. 2020; 12:CD004454. https://doi.org/10.1002/14651858.CD004454. pub4 PMID: 33368142
- The effect of antenatal steroids for fetal maturation on perinatal outcomes. NIH Consensus Statement. 1994; 12:1–24.
- Heindel JJ, Vandenberg LN. Developmental origins of health and disease: a paradigm for understanding disease cause and prevention. Curr Opin Pediatr. 2015; 27:248–253. <u>https://doi.org/10.1097/MOP.0000000000191</u> PMID: 25635586
- 6. Cottrell EC, Seckl JR. Prenatal stress, glucocorticoids and the programming of adult disease. Front Behav Neurosci. 2009: 3. https://doi.org/10.3389/neuro.08.019.2009 PMID: 19826624
- Braun T, Challis JR, John PN, Sloboda DM. Early-life glucocorticoid exposure: The hypothalamic-pituitary-adrenal axis, placental function, and long-term disease risk. Endocr Rev. 2013; 34:885–916. https://doi.org/10.1210/er.2013-1012 PMID: 23970762
- Kemp MW, Jobe AH, Usuda H, Nathanielsz PW, Li C, Kuo A, et al. Efficacy and safety of antenatal steroids. Am J Physiol Regul Integr Comp Physiol. 2018; 315:R825–R839. <u>https://doi.org/10.1152/ajpregu.00193.2017</u> PMID: 29641233

- Benediktsson R, Lindsay RS, Noble J, Seckl JR, Edwards CRW. Glucocorticoid exposure in utero: new model for adult hypertension. Lancet. 1993; 341:339–341. <u>https://doi.org/10.1016/0140-6736(93)</u> 90138-7 PMID: 8094115
- Drake AJ, Walker BR, Seckl JR. Intergenerational consequences of fetal programming by in utero exposure to glucocorticoids in rats. Am J Physiol Regul Integr Comp Physiol. 2005; 288:R34–R38. <a href="https://doi.org/10.1152/ajpregu.00106.2004">https://doi.org/10.1152/ajpregu.00106.2004</a> PMID: 15178540
- Nyirenda MJ, Welberg LA, Seckl JR. Programming hyperglycaemia in the rat through prenatal exposure to glucocorticoids-fetal effect or maternal influence? J Endocrinol. 2001; 170:653–660. https://doi.org/ 10.1677/joe.0.1700653 PMID: 11524246
- de Vries A, Holmes MC, Heijnis A, Seier JV, Heerden J, Louw J, et al. Prenatal dexamethasone exposure induces changes in nonhuman primate offspring cardiometabolic and hypothalamic-pituitary-adrenal axis function. J Clin Invest. 2007; 117:1058–1067. https://doi.org/10.1172/JCI30982 PMID: 17380204
- Sloboda DM, Moss TJM, Li S, Doherty DA, Nitsos I, Challis JRG, et al. Hepatic glucose regulation and metabolism in adult sheep: effects of prenatal betamethasone. Am J Physiol Endocrinol Metab. 2005; 289:E721–E728. https://doi.org/10.1152/ajpendo.00040.2005 PMID: 15928026
- Gatford KL, Wintour EM, De Blasio MJ, Owens JA, Dodic M. Differential timing for programming of glucose homoeostasis, sensitivity to insulin and blood pressure by in utero exposure to dexamethasone in sheep. Clin Sci (London, England). 2000; 98:553–560. https://doi.org/10.1042/cs0980553 PMID: 10781386
- Doyle LW, Ford GW, Davis NM, Callanan C. Antenatal corticosteroid therapy and blood pressure at 14 years of age in preterm children. Clin Sci (London, England). 2000; 98:137–142. PMID: 10657267
- Finken MJJ, Keijzer-Veen MG, Dekker FW, Frölich M, Walther FJ, Romijn JA, et al. Antenatal glucocorticoid treatment is not associated with long-term metabolic risks in individuals born before 32 weeks of gestation. Arch Dis Child Fetal Neonatal Ed. 2008; 93:F442–F447. https://doi.org/10.1136/adc.2007. 128470 PMID: 18450806
- Sacco A, Cornish EF, Marlow N, David AL, Giussani DA. The effect of antenatal corticosteroid use on offspring cardiovascular function: a systematic review. Int J Obstet Gynaecol. 2023; 130:325–333. https://doi.org/10.1111/1471-0528.17316 PMID: 36209465
- Darlow BA, Harris SL, Horwood LJ. Little evidence for long-term harm from antenatal corticosteroids in a population-based very low birthweight young adult cohort. Paediatr Perinat Epidemiol. 2022; 36:631– 639. https://doi.org/10.1111/ppe.12886 PMID: 35570644
- Washburn LK, Nixon PA, Snively BM, Russell GB, Shaltout HA, South AM, et al. Antenatal corticosteroids and cardio-metabolic outcomes in adolescents born with very low birth weight. Pediatr Res. 2017; 82:697–703. https://doi.org/10.1038/pr.2017.133 PMID: 28574979
- Dessens AB, Haas HS, Koppe JG. Twenty-year follow-up of antenatal corticosteroid treatment. Pediatrics. 2000; 105:e77. https://doi.org/10.1542/peds.105.6.e77 PMID: 10835090
- Liggins GC, Howie RN. A controlled trial of antepartum glucocorticoid treatment for prevention of the respiratory distress syndrome in premature infants. Pediatrics. 1972; 50:515–525. PMID: 4561295
- 22. Walters AGB, Lin L, Crowther CA, Gamble GD, Dalziel SR, Harding JE. Betamethasone for preterm birth: Auckland Steroid Trial full results and new insights 50 years on. J Pediatr. 2023; 255:80–88.e5. https://doi.org/10.1016/j.jpeds.2022.10.028 PMID: 36336005
- Dalziel SR, Walker NK, Parag V, Mantell C, Rea HH, Rodgers A, et al. Cardiovascular risk factors after antenatal exposure to betamethasone: 30-year follow-up of a randomised controlled trial. Lancet. 2005; 365:1856–1862. https://doi.org/10.1016/S0140-6736(05)66617-2 PMID: 15924982
- Dalziel SR, Rea HH, Walker NK, Parag V, Mantell C, Rodgers A, et al. Long term effects of antenatal betamethasone on lung function: 30 year follow up of a randomised controlled trial. Thorax. 2006; 61:678–683. https://doi.org/10.1136/thx.2005.051763 PMID: 16601084
- 25. Ministry of Health. Content Guide 2019/20: New Zealand Health Survey. Wellington: Ministry of Health; 2020. https://www.health.govt.nz/publication/questionnaires-and-content-guide-2019-20-new-zealand-health-survey.
- Price DB, Tinkelman DG, Nordyke RJ, Isonaka S, Halbert RJ, COPD Questionnaire Study Group. Scoring system and clinical application of COPD diagnostic questionnaires. Chest. 2006; 129:1531–1539. https://doi.org/10.1378/chest.129.6.1531 PMID: 16778271
- 27. Redfors B, Gregson J, Crowley A, McAndrew T, Ben-Yehuda O, Stone GW, et al. The win ratio approach for composite endpoints: practical guidance based on previous experience. Eur Heart J. 2020; 41:4391–4399. https://doi.org/10.1093/eurheartj/ehaa665 PMID: 32901285
- Ministry of Health. Annual Data Explorer 2021/22: New Zealand Health Survey [Data File]. 2022 [cited 2022 May 12]. https://minhealthnz.shinyapps.io/nz-health-survey-2021-22-annual-data-explorer/.

- Coppell KJ, Mann JI, Williams SM, Jo E, Drury PL, Miller JC, et al. Prevalence of diagnosed and undiagnosed diabetes and prediabetes in New Zealand: findings from the 2008/09 Adult Nutrition Survey. N Z Med J. 2013; 126:23–42.
- Crump C, Howell EA, Stroustrup A, McLaughlin MA, Sundquist J, Sundquist K. Association of preterm birth with risk of ischemic heart disease in adulthood. JAMA Pediatr. 2019; 173:736–743. <u>https://doi.org/10.1001/jamapediatrics.2019.1327 PMID: 31157896</u>
- Thompson JM, Mitchell EA, Borman B. Sex specific birthweight percentiles by gestational age for New Zealand. N Z Med J. 1994; 107:1–3. PMID: 8295745
- **32.** Ministry of Health. HISO 10001:2017 Ethnicity Data Protocols. Wellington, New Zealand: Ministry of Health; 2017.
- Cai X, Zhang Y, Li M, Wu JH, Mai L, Li J, et al. Association between prediabetes and risk of all cause mortality and cardiovascular disease: updated meta-analysis. BMJ. 2020; 370:m2297. <u>https://doi.org/ 10.1136/bmj.m2297</u> PMID: 32669282
- Lee SM, Shivakumar M, Park JW, Jung YM, Choe EK, Kwak SH, et al. Long-term cardiovascular outcomes of gestational diabetes mellitus: a prospective UK Biobank study. Cardiovasc Diabetol. 2022; 21:221. https://doi.org/10.1186/s12933-022-01663-w PMID: 36309714
- 35. Sebastian E, Bykersma C, Eggleston A, Eddy KE, Chim ST, Zahroh RI, et al. Cost-effectiveness of antenatal corticosteroids and tocolytic agents in the management of preterm birth: a systematic review. eClinicalMedicine. 2022: 49. https://doi.org/10.1016/j.eclinm.2022.101496 PMID: 35747187
- **36.** Reynolds RM. Glucocorticoid excess and the developmental origins of disease: two decades of testing the hypothesis– 2012 Curt Richter award winner. Psychoneuroendocrinology. 2013; 38:1–11. <u>https://doi.org/10.1016/j.psyneuen.2012.08.012 PMID: 22998948</u>
- Jobe AH, Goldenberg RL. Antenatal corticosteroids: an assessment of anticipated benefits and potential risks. Am J Obstet Gynecol. 2018; 219:62–74. <u>https://doi.org/10.1016/j.ajog.2018.04.007</u> PMID: 29630886
- McKinlay CJD, Cutfield WS, Battin MR, Dalziel SR, Crowther CA, Harding JE, et al. Cardiovascular risk factors in children after repeat doses of antenatal glucocorticoids: an RCT. Pediatrics. 2015; 135:e405– e415. https://doi.org/10.1542/peds.2014-2408 PMID: 25601978
- **39.** Doyle LW, Adams A-M, Robertson C, Ranganathan S, Davis NM, Lee KJ, et al. Increasing airway obstruction from 8 to 18 years in extremely preterm/low-birthweight survivors born in the surfactant era. Thorax. 2017; 72:712–719. https://doi.org/10.1136/thoraxjnl-2016-208524 PMID: 27601432
- 40. Crowther CA, Anderson PJ, McKinlay CJD, Harding JE, Ashwood PJ, Haslam RR, et al. Mid-childhood outcomes of repeat antenatal corticosteroids: a randomized controlled trial. Pediatrics. 2016: 138. https://doi.org/10.1542/peds.2016-0947 PMID: 27650051
- Ninan K, Liyanage SK, Murphy KE, Asztalos EV, McDonald SD. Evaluation of long-term outcomes associated with preterm exposure to antenatal corticosteroids: a systematic review and meta-analysis. Arch Gynecol Obstet. 2022; 176(6):e220483. https://doi.org/10.1001/jamapediatrics.2022.0483 PMID: 35404395
- **42.** Humbeck C, Jonassen S, Bringewatt A, Pervan M, Rody A, Bossung V. Timing of antenatal steroid administration for imminent preterm birth: results of a prospective observational study in Germany. Arch Gynecol Obstet. 2022. https://doi.org/10.1007/s00404-022-06724-9 PMID: 36042053
- Burvill AJ, Murray K, Knuiman MW, Hung J. Comparing self-reported and measured hypertension and hypercholesterolaemia at standard and more stringent diagnostic thresholds: the cross-sectional 2010– 2015 Busselton Healthy Ageing study. Clin Hypertens. 2022; 28:16. <u>https://doi.org/10.1186/s40885-022-00199-1</u> PMID: 35642010
- **44.** Xie D, Wang J. Comparison of self-reports and biomedical measurements on hypertension and diabetes among older adults in China. BMC Public Health. 2020; 20:1664. <u>https://doi.org/10.1186/s12889-020-09770-7 PMID: 33160325</u>